

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Leitner, Sandra M.; Stehrer, Robert

Working Paper Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms

wiiw Working Paper, No. 73

Provided in Cooperation with:

The Vienna Institute for International Economic Studies (wiiw) - Wiener Institut für Internationale Wirtschaftsvergleiche (wiiw)

Suggested Citation: Leitner, Sandra M.; Stehrer, Robert (2011) : Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms, wiiw Working Paper, No. 73, The Vienna Institute for International Economic Studies (wiiw), Vienna

This Version is available at: https://hdl.handle.net/10419/203938

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Wiener Institut für Internationale Wirtschaftsvergleiche The Vienna Institute for International Economic Studies

Working Papers | 73 | February 2011

Sandra M. Leitner and Robert Stehrer

Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms



wiiw Working Papers published since 2007

- No. 73 S. M. Leitner and R. Stehrer: Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms. February 2011
- No. 72 S. M. Leitner, J. Pöschl and R. Stehrer: Change begets change: Employment effects of technological and nontechnological innovations – A comparison across countries. January 2011
- No. 71 M. Holzner: Inequality, Growth and Public Spending in Central, East and Southeast Europe. October 2010
- No. 70 N. Foster, J. Pöschl and R. Stehrer: The Impact of Preferential Trade Agreements on the Margins of International Trade. September 2010
- No. 69 L. Podkaminer: Discrepancies between Purchasing Power Parities and Exchange Rates under the Law of One Price: A Puzzle (partly) Explained? September 2010
- No. 68 K. Hauzenberger and R. Stehrer: An Empirical Characterization of Redistribution Shocks and Output Dynamics. August 2010
- No. 67 R. Stöllinger, R. Stehrer and J. Pöschl: Austrian Exporters: A Firm-Level Analysis. July 2010
- No. 66 M. Holzner: Tourism and Economic Development: the Beach Disease? June 2010
- No. 65 A. Bhaduri: A Contribution to the Theory of Financial Fragility and Crisis. May 2010
- No. 64 L. Podkaminer: Why Are Goods Cheaper in Rich Countries? Beyond the Balassa-Samuelson Effect. April 2010
- No. 63 K. Laski, J. Osiatynski and J. Zieba: The Government Expenditure Multiplier and its Estimates for Poland in 2006-2009. March 2010
- No. 62 A. Bhaduri: The Implications of Financial Asset and Housing Markets on Profit- and Wage-led Growth: Some Results in Comparative Statics. February 2010
- No. 61 N. Foster and R. Stehrer: Preferential Trade Agreements and the Structure of International Trade. January 2010
- No. 60 J. Francois and B. Hoekman: Services Trade and Policy. December 2009
- No. 59 C. Lennon: Trade in Services: Cross-Border Trade vs. Commercial Presence. Evidence of Complementarity. November 2009
- No. 58 N. Foster and J. Pöschl: The Importance of Labour Mobility for Spillovers across Industries. October 2009
- No. 57 J. Crespo-Cuaresma, G. Doppelhofer and M. Feldkircher: The Determinants of Economic Growth in European Regions. September 2009
- No. 56 W. Koller and R. Stehrer: Trade Integration, Outsourcing and Employment in Austria: A Decomposition Approach. July 2009
- No. 55 U. Schneider and M. Wagner: Catching Growth Determinants with the Adaptive Lasso. June 2009
- No. 54 J. Crespo-Cuaresma, N. Foster and R. Stehrer: The Determinants of Regional Economic Growth by Quantile. May 2009
- No. 53 C. Lennon: Trade in Services and Trade in Goods: Differences and Complementarities. April 2009
- No. 52 J. F. Francois and C. R. Shiells: Dynamic Factor Price Equalization and International Convergence. March 2009
- No. 51 P. Esposito and R. Stehrer: Effects of High-Tech Capital, FDI and Outsourcing on Demand for Skills in West and East. March 2009
- No. 50 C. Fillat-Castejón, J. F. Francois and J. Wörz: Cross-Border Trade and FDI in Services. February 2009
- No. 49 L. Podkaminer: Real Convergence and Inflation: Long-Term Tendency vs. Short-Term Performance. December 2008
- No. 48 C. Bellak, M. Leibrecht and R. Stehrer: The Role of Public Policy in Closing Foreign Direct Investment Gaps: An Empirical Analysis. October 2008
- No. 47 N. Foster and R. Stehrer: Sectoral Productivity, Density and Agglomeration in the Wider Europe. September 2008
- No. 46 A. Iara: Skill Diffusion by Temporary Migration? Returns to Western European Work Experience in Central and East European Countries. July 2008
- No. 45 K. Laski: Do Increased Private Saving Rates Spur Economic Growth? September 2007
- No. 44 R. C. Feenstra: Globalization and Its Impact on Labour. July 2007
- No. 43 P. Esposito and R. Stehrer: The Sector Bias of Skill-biased Technical Change and the Rising Skill Premium in Transition Economies. May 2007

Sandra M. Leitner is economist at the Vienna Institute for International Economic Studies (wiiw). Robert Stehrer is wiiw Deputy Director of Research.

This publication was produced in the framework of MI-CRO-DYN (www.micro-dyn.eu), an international economic research project focusing on the competitiveness of firms, regions and industries in the knowledge-based economy. The project is funded by the EU Sixth Framework Programme (www.cordis.lu) with socio-economic sciences and humanities (http://ec.europa.eu/research/socialsciences/index_en.html).

This publication reflects only the authors' views, the European Community is not liable for any use that may be made of the information contained therein.





Sandra M. Leitner and Robert Stehrer

Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms

Contents

Abs	tract.		i			
1	Intro	duction	1			
2	Relat	ted literature and previous results				
<u>2</u> . 3.	Models of wage bargaining in closed and open economies					
	3.1	Bargaining in a closed economy	4			
	3.2	Bargaining in an open economy	6			
4.	Econ	ometric specification	7			
5.	The <i>i</i>	Austrian economy	10			
	5.1	The Austrian labor market – the system of collective wage bargaining	10			
	5.2	The Austrian economy – a general characterization	11			
6.	Data		13			
7.	Resu	Ilts	17			
	7.1	Exporters versus non-exporters	17			
	7.2	Across exporters with different exporter groups	20			
8.	Sum	mary and conclusion	24			
9.	Refe	rences	25			
10.). Appendix27					

List of Tables and Figures

Table 1	Sample overview, Austrian manufacturing: 2002-2006	14
Table 2	Relative shares of exporters, Austrian manufacturing: 2002-2006	14
Table 3	Firm-level indicators, weighted means 2002-2006, total manufacturing (NACE 15-37)	16
Table 4	Random effects model with 1-year lag, exporters versus non-exporters	18
Table 5	GMM estimation with 1-year lagexporters versus non-exporters	19
Table 6	Random effects model with 1-year lag. Across exporter groups	21
Table 7	GMM estimation with 1-year lagacross exporter groups	23
Table 8	Major trading partners for Austrian imports: import shares, 1980-2008	27
Table 9	Major trading partners for Austrian exports: export shares, 1980-2008	27
Figure 1	Real GDP and real openness	12
		12
Figure 2	and 2006	16

Abstract

This paper looks at the influence globalization exerts on wage negotiation processes and outcomes. Specifically, it establishes whether, compared to their purely domestically oriented counterparts, exporters share a higher fraction of the rents they generate with their employees. The analysis uses a panel of Austrian manufacturing firms between 2002 and 2006 and demonstrates that, in general, Austrian exporters do not share a higher part of their rents with their employees. Moreover, the analysis also takes into account that exporters are a very heterogeneous group, broadly differing in terms of the degree to which they trade internationally or to which they earn rents from their export activities. Against that backdrop, it determines whether rent-sharing systematically differs by the degree of internationalization of exporters. The results emphasize that particularly the most export-oriented firms are able to cut down on rent-sharing which corroborates the idea that exporters can credibly and effectively exploit their threat-points of either outsourcing or off-shoring part of their production which induces employees to concede to more moderate wage changes so as to avert the potential loss of employment.

Keywords: wage determination, rent-sharing, internationalization, firm-level analysis

JEL classification: F16, J31, L6

Sandra M. Leitner and Robert Stehrer

Do exporters share part of their rents with their employees? Evidence from Austrian manufacturing firms

1. Introduction

Theoretically, the observable growing global integration of economies has strong potentials to fundamentally alter wage negotiation processes and dynamics between employers and employees and to reframe and change the associated wage outcomes.

Firstly, globalization can be decisive for the relative bargaining position of employees and employers. It may initiate a redistribution of *bargaining power* to the employers' advantage thereby harming employees' bargaining position which limits the wage claims they can realize and the share of rents they can siphon off.

Secondly, globalization may directly affect the *threat-points* both employees and employers have at their disposal during wage negotiations. On the one hand, employees may use unemployment benefits to temporarily tide over unemployment spells or find alternative employment in other firms should wage negotiations break down. These outside options provide them with a strong position during negotiations. On the other hand, employers may look for alternative outside options and may threaten their employees to source from abroad (outsourcing) or even move production abroad (offshoring) should they fail to concede, thereby depriving them of their jobs. These threats are particularly credible if realistic and feasible outside options exist. Moreover, they are especially effective should employees lack attractive fallback positions which further strengthens employers' bargaining positions. However, employees may also benefit from associated hold-up effects of imports since imports of intermediates need to be specified in advance. This buys employees time and provides them with sufficient leverage during wage negotiations to claim a share of their employers' rents.

Finally, globalization can also impinge on the *size of profits* firms are able to realize in light of intense global competition. And smaller or shrinking profits are expected to reduce employers' willingness to share and to limit the degree of rent-sharing between employers and employees.

And while the theoretical links between globalization, wage negotiation processes and rentsharing are well established, empirical evidence is still meager. This study contributes to the slowly burgeoning literature on rent-sharing in open economies by analyzing the prevalence and degree of rent-sharing among Austrian exporters between 2002 and 2006. Generally, exporters which operate internationally are better informed about potential outsourcing or offshoring options - their threat-points – and can therefore apply them during wage negotiations to limit or reduce the degree of rent-sharing employees bargain for. Hence thematically, the current study contributes to the second channel between globalization and wage outcomes outlined above.

The analysis shows that, in general, exporters do not share part of their rents with their employees. This finding is however qualified if a more refined classification scheme is applied that accounts for the strong heterogeneity among exporters. It demonstrates that only the most export-oriented exporters which earn more than 50 percent of their sales from their export activities can effectively exploit their threat-points of outsourcing or offshoring which translates into lower rent-sharing between employees and their employers. Moreover, evidence is found that larger and more skill or technology-intensive firms also pay extra premia to their employees.

The remainder of the paper is structured as follows: section 2 provides a brief discusses of related empirical results. Section 3 outlines wage-bargaining models in closed and open economies to demonstrate how globalization may affect the wage determination process theoretically. The methodological approach is highlighted in section 4 while section 5 highlights why Austria is an interesting case to study. Section 6 discusses the data used in the empirical analysis. The results are presented and discussed in section 7 while, finally, section 8 summarizes and concludes.

2. Related literature and previous results

Whether workers are able to appropriate some of their employers' surplus has been a hotly debated and controversial issue among economists for decades. Theoretically, the question is of enormous explosiveness as considerable doubt is cast on the validity of conventional competitive models to properly represent labor markets should evidence of rentsharing manifest. Consequently, on economic-political grounds, it is of great significance as a correction of macroeconomic and labor market policies becomes inevitable. In general, empirical analyses find a robust and significant positive correlation between profitability and wages which is taken as conclusive evidence that employers indeed share part of their rents with their employees.

Following the seminal work of Slichter (1950) for the US manufacturing sector, a rich strand of literature strongly confirms that industry profitability is of significant importance for industry wage determination. For example, Dickens and Katz (1986) or Katz and Summers (1989) for the US, Kouwenberg and van Opstal (1999) for the Netherlands or Benito (2000) for the UK demonstrate that more profitable industries pay higher wages. In the same vein, Estevao and Tevlin (2000) confirm that the lion's share of observed wage dispersion in the US manufacturing sector results from industry financial conditions.

More recently, the widespread availability of firm-level data rendered the estimation of more direct measures of rent-sharing between employers and employees possible and paved the way for a rich strand of literature that again consistently confirms that workers' wages are not independent of diverse measures of employer's ability to pay. This wage-ability-to-pay nexus is typically expressed in terms of the Lester range which captures the change in wages should employees hypothetically switch jobs from low to a high-rent firms¹. In that respect, Blanchflower et al. (1996) highlight that the mere switch of jobs of employees from a low-profit to a high-profit US firm would increase their wages by about 24 percent. Similarly, Hildreth and Oswald (1997) use small sets of UK company and establishment data to emphasize that employees would experience a 16 percent rise in wages should they move from low to high-profit entities. In contrast, a wage-surge of around 60 percent is reported by Goos and Konings (2000) should employees shift from low to high-profit Belgian firms. Moreover, that unions can capture approximately 20 percent of total quasi-rents is emphasized by Abowd and Lemieux (1993) in their study on Canadian contract data.

However, empirical analyses based on either industry or firm-level data came under severe criticism for their lack of controlling for worker characteristics. Remedy came in the form of matched employer-employee data which become available for an increasing number of countries. With the exception of France (Margolis and Salvanes, 2001), numerous empirical studies consistently demonstrate that rent-sharing plays a central role in wage determination and that it is of comparable magnitude across countries considered (see e.g. Arai (2003) for Sweden, Hildreth (1998) for the UK, Martins (2003) for Portugal or Piekkola and Kauhanen (2003) for Finland). These countries are characterized by different institutional settings or bargaining structures. For example, bargaining is least centralized in Canada, the US or more recently also the UK, intermediate in continental European countries and traditionally most centralized in Nordic countries (OECD, 2004). Consequently, above results appear to suggest that the phenomenon of rent-sharing is independent of prevailing institutional bargaining structures. Additionally, these studies also shed light on the significance of other firm characteristics for wage determination or rent-sharing processes. They emphasize that workers typically benefit from working in larger, more R&D intensive or innovative, more capital-intensive firms or in firms whose owners have invested in new process technologies.

Surprisingly, however, even though the last decades witnessed growing global integration of economies in terms of foreign direct investments (FDI), trade, migration or the diffusion of technology, none of the above-mentioned studies explicitly accounted for the effects of globalization on labor markets in general or rent-sharing in particular. But the degree of internationalization of activities is pivotal to employees' and employers' bargaining posi-

¹ The Lester range is defined as the elasticity of wages with respect to profits multiplied by four times the ratio between the standard deviation of profits and mean profits.

tions during wage negotiations which in turn determines the extent of rent-sharing between them. Specifically, the idea that globalization may directly affect employees' *bargaining power* is taken up by Brock and Dobbelaere (2006). They use a panel of Belgian manufacturing firms and provide first empirical evidence that neither intense import competition, strong export-orientation, outsourcing of the production of intermediates or high inward FDI exert much impact on employees' bargaining power. Surprisingly though, tariffs which protect domestic markets from foreign competition and are expected to strengthen employees' bargaining position turn out to have a negative impact, significantly reducing their ability to siphon off a share of their employers' rents.

Furthermore, globalization may also affect firm's and workers' *threat-points* since employers may threat to source from abroad or move production abroad, depriving workers of their jobs or outside employment options. In contrast, workers may benefit from hold-up effects of imports when the need to pre-specify intermediate imports buys workers sufficient time to negotiate for a share of rents realized by their employers. In a study on French employer-employee data Kramarz (2003) shows that the positive hold-up effect is more than compensated by the negative threat-effect of sourcing or producing abroad, leaving most employees which work in strongly import-oriented industries with on average lower wages. Similarly, Choi (2001) focuses on the effect outward FDI has on union wage premia in the US manufacturing sector. He finds evidence of an effective threat-effect such that the fear of losing their jobs tends to induce employees to accept lower shares of their employers' rents.

Finally, globalization and with it more intense competition may affect the size of the rents firms are able to make and employees are able to appropriate. Abowd and Lemieux (1993) use Canadian contract data and highlight that increased foreign competition, for example in terms of lower import prices, reduces both wages as well as quasi-rents per worker.

Apparently, empirical evidence on rent-sharing in a global economy is rather scarce, a shortcoming the ensuing analysis seeks to tackle.

3. Models of wage bargaining in closed and open economies

3.1 Bargaining in a closed economy

In what follows, a Right-to-Manage (RTM) approach is applied that strongly rests on the model developed by Blanchflower et al. (1996). In contrast to an efficient bargaining approach (EB), the Right-to-Manage approach is more suitable to the Austrian case since during wage negotiations workers bargain over wages or working hours only but have no say in determining the overall level of employment.

Specifically, at time *t*, firm *i* uses the following production technology to generate its output Y_{it} :

$$Y_{it} = A_{it}f(n_{it}, k_{it}), \qquad (1)$$

where A_{it} captures total factor productivity, n_{it} is total employment and k_{it} refers to capital input. In the presence of short-run capital quasi-fixity, firm *i* equates its utility to its shortrun profits π_{it} , where $\pi(w_{it}, n_{it}) = R(n_{it}) - w_{it}n_{it}$ and w_{it} is the wage. $R_{it} = p_{it}Y_{it}$ refers to total revenues, p_{it} is the output price and Y_{it} is total output. Above specification implies that if no consensus is reached during collective wage negotiations, firm *i*'s profits turn out to be zero as workers refuse to work.

The trade union is assumed to be risk-neutral. It represents the interests of a pool of homogenous members and has a 'utilitarian' objective in that it maximizes the utility of all its members as captured by the following objective function:

$$U = [u(w_{it}) - u(\overline{w}_{it})]n_{it}$$
⁽²⁾

where $u(w_{it})$ is a single union member's utility from wage income w_{it} and \overline{w}_{it} is the wage available from alternative employment in the event of a breakdown of collective wage negotiations and represents the union's single threat-point.

In the course of collective bargaining, wages are determined by maximizing the following Nash bargaining problem:

$$N = \max \theta \log[(u(w_{it}) - u(\overline{w}_{it}))n_{it}] + (1 - \theta)\log[\pi_{it}], \qquad (3)$$

where $\theta \in [0, 1]$ refers to the bargaining power of employees while $1 - \theta$ captures employers' bargaining power.

The maximization of N with respect to the wage rate w_{it} gives the familiar expression:

$$\boldsymbol{w}_{it} \cong \overline{\boldsymbol{w}}_{it} + \left(\frac{\theta}{1-\theta}\right) \left(\frac{\pi_{it}}{n_{it}}\right). \tag{4}$$

Hence, equilibrium wages are determined by workers' bargaining power θ , the 'outside' wage \overline{w}_{it} and a firm's profits per employee. Following Blanchflower et al. (1996), the 'outside' wage \overline{w}_{it} is typically assumed to be a function $b(w_{it}^o, UB_t, U_t)$ of three determining factors: the going wage rate w_{it}^o in other sectors of the economy, the level of unemployment benefit *UB* available to the temporary unemployed as well as the economy-wide unemployment rate *U*. The significance of these factors is corroborated by Christofides and Oswald (1992) who analyze the effect of pressures from the outside market on wage determination. They find robust empirical evidence that both the going wage rate w_{it}^o as well as the level of unemployment benefit *UB* have a positive and significant effect on wages but that higher rates of unemployment *U* are associated with lower wages.

Hence, above expression becomes:

$$w_{it} \simeq b(w_{it}^{o}, UB_{t}, U_{t}) + \left(\frac{\theta}{1-\theta}\right) \left(\frac{\pi_{it}}{n_{it}}\right).$$
(5)

3.2 Bargaining in an open economy

However, with globalization and an opening-up of domestic markets to international activities, firms are no longer confined to purely domestic factors or conditions. More specifically, instead of sourcing production factors or intermediate goods from domestic suppliers or producing them themselves, firms may now alternatively source from abroad (outsourcing). Moreover, firms may now relocate part or all of their production abroad to exploit economically more favorable foreign conditions (offshoring). Both options of either outsourcing or offshoring crucially affect wage negotiation processes between employers and employees. In contrast to the closed-economy scenario in which a breakdown of wage negotiations leaves firms with zero profits, firms operating in open economies can take advantage of outside options and therefore still realize positive profits even if no consensus is reached during wage negotiations. Hence, provided that the threat of outsourcing or offshoring can be credibly made, the emerging improved fallback position strengthens employers' position during wage negotiations, limiting the share of rents employees may be able to appropriate.

Outsourcing: Following Kramarz (2003), if negotiations break down and production is stopped, firms can still import intermediate goods I_{it} and employ temporary workers n_{it}^{tmp}

to generate the following profits:

.....

$$\pi_{it}' = R(n_{it}^{imp}, I_{it}) - w_{it}^{imp} n_{it}^{imp} - p_{it}' I_{it}$$
(6)

where w_{it}^{tmp} refers to temporary workers' wage and p_{it}^{l} denotes the price of imported intermediates².

Offshoring: On the other hand, following Rodrik (1999), production can be moved abroad to realize profits π_{it}^* with foreign employment n_{it}^* at the going wage rate w_{it}^* :

$$\pi_{it}^{*} = R(n_{it}) - w_{it}n_{it}^{*}$$
(7)

Both outside options, π_{it}^{l} or π_{it}^{*} become crucial threat-points during wage negotiations which can be exploited to the employer's advantage. With π_{it}^{o} denoting a firm's profits from utilizing its outside options (i.e. outsourcing or offshoring), an alternative Nash bargaining problem can be specified

$$N = \max \theta \log[(u(w_{it}) - u(\overline{w}_{it}))n_{it}] + (1 - \theta)\log[\pi_{it} - \pi_{it}^{o}]$$
(8)

² According to Blanchflower et al. (1996), outside wages for temporary workers are defined as: w_{it}^{o} (see above).

that gives the following solution for wages in an open-economy scenario:

$$w_{it} \cong b(w_{it}^{o}, UB_{it}, U_{it}) + \left(\frac{\theta}{1-\theta}\right) \left(\frac{\pi_{it} - \pi_{it}^{o}}{n_{it}}\right).$$
(9)

This implies that

$$\frac{\partial w_{it}}{\partial \pi_{it}^o} < 0 \tag{10}$$

Hence, equation (9) highlights that the more profitable are the outside options of employers π_{it}^{o} and the more credible are employers' threats to actually realize them, the lower will be the wage w_{it} agreed upon during wage negotiations.

4. Econometric specification

Not all firms operating in open economies can credibly threaten their employees with either outsourcing or offshoring part of their production, though. Firms that already move on the international scene and intensely export to international markets, import from foreign suppliers or already produce abroad are generally better informed about their outside options. They are usually more familiar with alternative sources or suppliers for their intermediate goods and more knowledgeable about prevailing foreign production capabilities and capacities which could be harnessed should production be relocated abroad. In contrast, purely domestically oriented firms which source, produce and supply domestically only usually lack comparable information on potential outside options or fallback positions which severely undermines the credibility of their threats and drastically weakens their bargaining position. Hence, in general, firms which operate internationally can more effectively exploit their threat-points which, in turn, significantly strengthens their bargaining position and limits or lowers the rents they share with their employees.

The ensuing analysis is particularly interested in the rent-sharing behavior of Austrian exporters which earn from exporting tradable goods and services to different degrees. Martins and Opromolla (2010) for Portuguese firms as well as Bernard et al. (2005) in their study on US-based firms emphasize that a firm's export status strongly correlates with its import status, rendering the export status of a firm a good proxy for its overall international trade activities. Hence, the analysis implicitly also accounts for firms' imports activities.

From equation (9) in section 3, the following econometric specification is derived:

$$\boldsymbol{w}_{ijt} = \alpha_i + \theta \left(\frac{R}{n}\right)_{ijt} + \beta \boldsymbol{X}_{ijt} + \gamma \boldsymbol{E} \boldsymbol{X} \boldsymbol{P}_{ijt} + \delta \boldsymbol{E} \boldsymbol{X} \boldsymbol{P}_{ijt} \times \left(\frac{R}{n}\right)_{ijt} + \alpha_j + \alpha_t + \varepsilon_{ijt}, \qquad (11)$$

where w_{ijt} is the log of nominal gross wage costs per employee in firm *i* and industry *j*. The wage costs are the sum of all gross wages and salaries of all manual and non-manual workers, respectively (including gross compensations for all apprentices and homeworkers) plus all legal social security contributions and other social expenditures of the employer.

 $(R/n)_{ijt}$ is the log of labor productivity defined as sales per employee and is used as a proxy for a firm's financial position, or equivalently, its ability-to-pay. θ is the rent-sharing coefficient the analysis seeks to identify.

 \mathbf{X}_{ijt} captures additional firm characteristics like size, R&D efforts or technology intensity. Firm size is proxied by the log of the number of employees and included to capture potential size-wage premia. Empirical results as to the significance of size-wage premia are mixed and inconclusive, however. And while Arai (2003), Fakhfakh and FitzRoy (2004) or Piekkola and Kauhanen (2003) for Swedish, French and Finnish matched employeremployee datasets stress that, on average, larger Swedish, French or Finnish firms, respectively, pay higher wages to their workforces, Korösi (2010) shows that a significant negative size-wage effect prevailed for a representative sample of Hungarian firms in the period spanning from 1998 to 2006. Furthermore, no significant effect emerges in the study of Hildreth and Oswald (1997) for a small and unbalanced UK firm panel of about 300 firms.

R&D efforts enter in terms of the ratio of R&D personnel to overall firm employment and are included to account for a firm's innovative efforts aimed at developing productivityenhancing product and/or process innovations that warrant temporary monopoly positions and allow for a temporary siphoning off of above-normal monopoly profits. These extraordinary profits may be shared with the firm's personnel, irrespective of skill-level, which then translates into higher wages. This positive effect is corroborated by the studies of Korösi (2010) for a sample of Hungarian firms and Piekkola and Kauhanen (2003) for a sample of Finnish matched employer-employee data which both highlight that innovative firms indeed pay an extra premium to their workforce. The latter study furthermore highlights that more educated employees receive a higher premium, on average, than less educated ones.

A firm's technology intensity is captured by the log of total software expenditures per employee and is used as a control variable for worker characteristics that are generally absent in the available firm-level data set. Specifically, following Bronars and Famulari (2001) it can be argued that if technology and skilled labor are strategic complements, technology intensive firms hire workers with higher observable and unobservable skills. Hence, if left unaccounted for, the rent-sharing coefficient might simply be picking up the effects of higher unobserved ability, biasing the rent-sharing coefficient upwards. Empirical analyses conducted with matched employer-employee data by Arai (2003) for Sweden, Fakhfakh and FitzRoy (2004) for France, Martins (2003) for Portugal or with firm-level data by Brock and Dobbelaere (2006) for Belgium demonstrate that a firm's capital intensity has a significant positive effect on average firm wages which can be taken as conclusive evidence that high skilled employees indeed sort into high-technology, high-wage firms. However, with the exception of Arai (2003), all of the above-mentioned studies refute the Bronars and Famulari hypothesis as the rent-sharing coefficients remain unresponsive to the inclusion of a firm's capital-intensity proxy.

EXP_{iit} is a proxy for a firm's export status, included in terms of export dummies. The ensuing analysis looks at two different export classification schemes: a general scheme that differentiates between exporters with positive export shares on the one hand and nonexporters with zero export shares on the other. Hence, it can be established whether, on average, exporters pay higher wages than non-exporters which might be necessary if workers demand premiums due to the potentially riskier or more demanding activities performed in exporting firms. However, exporters are a very heterogeneous group so that lumping them together might dwarf or obscure more disaggregated effects. Hence, the analysis also seeks to show whether wages systematically differ by the degree of internationalization of firms. Therefore, a more refined scheme is used that classifies exporters into four different groups, depending on the reported export shares: firms with export shares of less or equal to 5 percent, firms with export shares of more than 5 but less or equal to 30 percent, firms with export shares between 30 and 50 percent and finally, firms with reported export shares exceeding 50 percent. Previous empirical studies find conclusive evidence of a significant positive export premium, even after controlling for observed and unobserved firm and worker characteristics (see e.g. Martins and Opromolla (2010) for Portugal, Munch and Skaksen (2008) for Denmark, Schank et al. (2007) for Germany or Tsou et al. (2006) for Taiwan).

Furthermore, interaction terms $EXP_{ijt} \times (R/n)_{ijt}$ between a firm's export status and labor productivity are included to capture whether rent-sharing is more or less pronounced in exporting firms as compared to non-exporting firms or whether the extent of rent-sharing systematically differs across exporting firms with different export shares. Generally, the expected effects are unclear a priory. On the one hand, exporting firms may exploit the threat-point they gain from the internationalization of their operations and the potential change in their sourcing strategies of intermediate or finished products. Specifically, instead of sourcing and producing domestically, they can either source from abroad or even move part of their production process abroad. In that case, foreign products become close substitutes for domestic workers, weakening workers' bargaining position. Moreover, this effect is compounded when due to foreign sourcing strategies workers' outside options deteriorate. On the other hand, workers' bargaining position may improve due to the positive hold-up effect that emerges if intermediate products need to be specified and purchased in advance. Some empirical evidence is provided by Kramarz (2003) for a unique French matched employer-employee dataset. All in all, he finds a negative wage effect for most workers, with the negative effect of outsourcing outweighing any positive hold-up effect.

Moreover, a firm's sourcing strategy may vary with the degree of internationalization, as captured by its export share. Specifically, more export-oriented and international firms may have a higher willingness to source from abroad or to move production abroad, which strengthens their threat-points and potentially reduces their willingness to share rents.

The right-to-manage rent-sharing model discussed above (see equations (5) and (9)) also highlights the role attributed to the going wage rate w_{it}^{o} , the level of unemployment benefit *UB* as well as the overall unemployment rate *U* in determining equilibrium wages. And while the former is captured by a set of 2-digit industry dummies (α_j), the latter two are controlled for by year dummies (α_t).

Finally, firm fixed effects are included (α_i) to account for unobserved firm heterogeneity which may systematically affect firm wages or wage policies. Such unobserved firm characteristics could be related to very specific production processes or systems of work organization. Specifically, production processes that involve dangerous work which necessitate the payment of compensation or risk premia but also provide the firm with higher profits or a specific organization of work which rests on the payment of efficiency wages instead of active supervision to spur productivity and consequently profitability may systematically bias the rent-sharing coefficient upwards (Margolis and Salvanes, 2001).

In what follows, a brief description of the Austrian collective bargaining system as well as a general characterization of the Austrian economy and her development during the last three to four decades is given as a thematic frame for the ensuing empirical analysis which will follow in section 7 below.

5. The Austrian economy

5.1 The Austrian labor market – the system of collective wage bargaining

In Austria, with a predominantly sectoral wage bargaining system, a relatively centralized system is in place (Du Caju et al., 2008). Every year, about 400 collective agreements are reached between the different sector-specific trade associations of the Austrian Federal Economic Chamber (Wirtschaftskammer Österreich), as representatives of employer interests on the one hand, and sectorally organized trade unions, as representatives of employee interests, on the other. These agreements are generally valid throughout Austria but in some sectors, like the food, beverage and tobacco industry, collective bargaining is organized regionally, leading to regionally differentiated agreements. Furthermore, wage

bargaining also takes place at the occupational level, separately for manual and nonmanual workers. Generally, collective wage negotiations in Austria are highly consensual, as evidenced by the historically low strike incidence. Specifically, since 2005, no single strike has occurred in Austria (WKÖ, 2009).

The average length of collective bargaining agreements is one year. Wage negotiations are highly seasonal, peaking in January and in May, and are strongly coordinated across the economy due to the practice of 'pattern bargaining'. Specifically, annual collective wage negotiations typically start with the metal industry in November, whose outcomes set a benchmark for ensuing negotiations and therefore have a leading role in the overall annual bargaining process (Knell and Stiglbauer, 2009). And like in some Nordic countries, minimum wages in Austria are part and parcel of collective wage agreements and are negotiated separately for each sector. During wage negotiations, trade unions traditionally demand the compensation of the inflation rate plus part of the productivity increases.

Just like in many other OECD countries, trade union density, defined as the percentage of workers who are members of a trade union, continuously declined in Austria from 63 percent in 1970 to only 37 percent in 2000 (OECD, 2004). However, despite the declining trade union density, trade union coverage is extraordinarily high at about 98 to 99 percent (OECD, 2004). This apparent discrepancy may be explained by the widespread use of extension procedures for wage agreements or the voluntary application of wage agreements to non-union members, rendering trade-union membership less attractive or less advantageous.

5.2 The Austrian economy – a general characterization

Austria is considered an economically stable small and very open economy with relatively low and stable inflation rates ranging between 2 to 4 percent annually since the 1980s and low unemployment rates with on average 4 percent per year between 1990 and 2009.

Austria is one of the richest countries, ranking 11th in 2009 worldwide and 4th in 2009 in the EU (in terms of GDP per capita).³ And except for the slump in 1993 and the more pronounced recession between 2001 and 2003, over the last three decades, the Austrian economy has steadily been growing by about 2 percent annually, on average (Figure 1). And between 1988 and 2009, Austrian real GDP almost doubled from about 160 to approximately 260 Billion Euros.

As a small economy, Austria is very outward-oriented and open to international business relations in order to compensate for any technological and endowment-specific shortcom-

³ World Economic Outlook Database-April 2010, International Monetary Fund.

ings. In that respect, real openness, defined as the sum of real exports and imports over real GDP, was 70 percent in 1995. And only five years later, in 2000, it already reached 90 percent. In 2004 it finally crossed the 100 percent line and peaked at 113 percent in 2007, before it plunged below 100 percent in 2009 due to the global recession (Figure 1).



Figure 1

Real GDP and real openness

Source: own calculations

In terms of the source of Austrian imports or the destination of Austrian exports, multilateral trade relations are deep seated within the European Union as more than 60 percent of both Austrian exports and imports are destined for or originate from within the European Union. More specifically, from 1980 to 2008, the share of Austrian imports coming from members of the European Union increased from about 68 percent to 78 percent (see Table 8 in the Appendix). Furthermore, within the same period, the share of Austrian exports destined for economies within the European Union grew from 67 percent in 1980 to 72 percent in 2008 (see Table 9 in the Appendix).

Historically, Germany has been Austria's major trading partner. Specifically, between 40 to 46 percent of Austrian imports originate from Germany while between 30 to 39 percent of all Austrian exports are destined for the German market. Furthermore, Italy has been Austria's second most important trading partner, providing around 7 to 9 percent of her imports and absorbing approximately 10 percent of her exports. Moreover, Austria obtains 4 to 5 percent of her imports from Switzerland and the Netherlands and around 3 percent from France and the Czech Republic. The latter has become an increasingly important country of origin for Austrian imports. Specifically, between 1995 and 2008, the share of imports originating in the Czech Republic doubled. Finally, imports from China almost quadrupled from only around 0.7 percent in 1980 to around 2.5 percent in 2008 while the US has lost

ground as the share of imports from the US has halved from around 4 percent in the 1990s to only 2 percent in 2000. Moreover, particularly over the last decade, the relative importance of Austria's major export markets has shifted and the New Member States have become increasingly more important, not just in terms of exports, but also in terms of imports. And given the relatively low labor costs, neighboring New Member States have become attractive destinations for outsourcing or offshoring activities of Austrian firms.

Finally, Austria's main import products are machinery and equipment, motor vehicles, chemicals, metal goods, oil and oil products and foodstuffs and her main export products are machinery and equipment, motor vehicles and parts, paper and paperboard, metal goods, chemicals, iron and steel, textiles and foodstuffs.

6. Data

The analysis uses data provided by Statistics Austria via 'remote execution'. The core dataset is the Austrian 'Performance and Structure Survey' (Leistungs- und Strukturerhebung) for the period 1997-2006 and NACE categories C to F.⁴ The current analysis uses data for the manufacturing sector (NACE D) only and, due to some fundamental methodological changes in 2002, focuses on the period from 2002 to 2006. The dataset provides annual firm-level information on numerous performance and structure indicators like the production value, sales, gross value-added, employment, total investment, R&D expenditures, R&D employment and wages and salaries, among other things.

However, the 'Performance and Structure Survey' does not collect information on a firm's export behavior and status. Therefore, the 'Performance and Structure Survey' was matched with the Austrian 'Business and Consumer Survey' (Konjunkturstatistik) which is a monthly survey that, apart from domestic sales, also collects information on export sales. A firm's export sales (as a share of overall sales) were used to derive export indicator variables or export markers which allow for distinguishing between non-exporters and exporters in general and exporters with different export shares in particular. Particularly, a firm was considered to be an exporter if in any particular year it reported positive export sales, while zero export sales were taken as an indication of a non-exporter. Furthermore, depending on their export shares (defined as export sales as a percentage of overall firm sales), exporters were categorized into four groups: 'marginal exporters' with export shares of equal to or less than 5 percent, 'low to medium exporters' with export shares of more than 5 but less or equal to 30 percent, 'high exporters' with export shares between 30 and 50 percent, and finally, 'very high exporters' with export shares exceeding 50 percent.

⁴ In 1996, The Austrian Performance and Structure Survey replaced its predecessor, the Austrian Industry Statistics Survey (Industriestatistik), and was first conducted by Statistics Austria in 1997.

Table 1

Sample overview, Austrian manufacturing: 2002-2006

Year	Total number of firms	Firms with known export status	Exporters	Non-Exporters	Share of exporters (in %)
2002	27,572	5,973	3,218	2,755	53.88
2003	28,581	6,054	3,303	2,751	54.56
2004	28,609	5,949	3,340	2,609	56.14
2005	28,374	5,719	3,248	2,471	56.79
2006	28,712	6,326	3,537	2,789	55.91

Exporter groups are defined as follows: 'marginal exporters' with export shares of equal to or less than 5 percent, 'low to medium exporters' with export shares of more than 5 but less or equal to 30 percent, 'high exporters' with export shares between 30 and 50 percent, and finally, 'very high exporters' with export shares exceeding 50 percent.

Source: own calculations

Table 2

Relative shares of exporters, Austrian manufacturing: 2002-2006

							Average
NACE	Industry	2002	2003	2004	2005	2006	growth rates
15	Food and beverages	23.60	24.31	25.48	26.67	27.93	4.30
16	Tobacco products	-	-	-	-	-	-
17	Textiles	84.38	86.84	85.53	85.23	85.42	0.32
18	Wearing apparel	74.49	78.89	77.22	77.03	75.61	0.43
19	Leather	86.11	88.46	87.50	85.71	83.33	-0.79
20	Wood	50.00	53.23	53.14	54.71	50.99	0.61
21	Pulp and paper	87.95	89.02	88.61	90.79	91.46	0.99
22	Publishing and printing	59.18	59.91	64.82	65.99	66.37	2.95
23	Refined petroleum	66.67	50.00	50.00	50.00	50.00	-6.25
24	Chemicals	86.96	92.04	91.07	89.29	89.76	0.84
25	Rubber and plastic products	85.27	87.32	89.25	87.92	89.69	1.28
26	Non-metallic mineral products	37.91	40.95	40.06	41.62	41.54	2.39
27	Basic metals	96.77	98.94	97.83	95.79	97.87	0.30
28	Fabricated metal products	51.80	51.66	53.73	54.36	53.84	0.99
29	Machinery and equipment	84.25	87.95	89.58	89.63	89.61	1.57
30	Office machinery and computers	83.33	62.50	77.78	66.67	66.67	-3.71
31	Electrical machinery	83.72	83.45	84.17	85.61	80.56	-0.91
32	Radio, TV, communication	93.02	90.70	89.36	93.48	90.20	-0.72
33	Precision & optical instruments	50.26	48.57	53.14	53.96	49.22	-0.30
34	Motor vehicles	83.78	86.84	87.95	90.36	89.13	1.58
35	Other transport equipment	93.75	88.89	89.47	85.71	85.71	-2.18
36	Manufactures n.e.c.	43.32	44.16	45.53	44.50	39.43	-2.15
37	Recycling	77.78	78.95	80.95	80.95	80.77	0.95
15-37	Total manufacturing	53.88	54.56	56.14	56.79	55.91	0.94
Source:	own calculations						

Due to the smaller sample size of the 'Business and Consumer Survey' and restrictive confidentiality requirements that leave cells with less than 4 observations unaccounted for, the overall sample size of 27,572 in 2002 and 28,712 in 2006 was reduced to between 5,973 in 2002 and 6,326 in 2006 (Table 1). Between 2002 and 2006, the share of exporters in the sample remained relatively stable, with only a slight increase from 54 percent in 2002 to 56 percent in 2006.

Table 2 reports the relative shares of exporters by two-digit NACE category for the period 2002 to 2006. On average, between 2002 and 2006, about 55 percent of all firms in manufacturing were exporters. A closer look at the more disaggregated 2-digit level furthermore reveals that only 3 out of 23 2-digit industries are characterized by relative exporter shares below 50 percent: Food and beverages (NACE 15), Non-metallic mineral products (NACE 26) and Manufactures n.e.c. (NACE 36). On the other hand, in 12 out of 23 2-digit industries, more than 80 percent of all firms are exporters: Textiles (NACE 17), Leather (NACE 19), Pulp and paper (NACE 21), Chemicals (NACE 24), Rubber and plastic products (NACE 25), Basic metals (NACE 27), Machinery and equipment (NACE 29), Electrical machinery (NACE 31), Radio, TV, communication (NACE 32), Motor vehicles (NACE 34), Other transport equipment (NACE 35), Recycling (NACE 37). Finally, between 2002 and 2006, the three industries which experienced the highest average increases in the relative number of exporters are Food and beverages (NACE 15), Publishing and printing (NACE 22) and Non-metallic mineral products (NACE 26). In contrast, the three industries with the highest losses in the relative number of exporters are Refined petroleum (NACE 23), Office machinery and computers (NACE 30) and Other transport equipment (NACE 35).

A more refined and detailed representation of the share of exporters in each 2-digit NACE industry is provided in Figure 2 which depicts relative shares of exporters by exporter groups, differentiating between 'marginal exporters', 'low to medium exporters', 'high exporters', and 'very high exporters'. Apparently, most industries are dominated by exporters with export shares of more than 5 but less or equal to 30 percent (i.e. 'low to medium exporters') on the one hand, and of export shares exceeding 50 percent (i.e. 'very high exporters'), on the other.

Finally, Table 3 presents weighted means of the overall set of variables used throughout the empirical analysis, separately for exporters and non-exporters as well as for the four exporter-groups considered. It shows that in general, exporters outperform non-exporters. Specifically, in contrast to non-exporters, exporters pay higher wages per employee, have higher labor productivities, report higher software expenditures per employee, employ a higher fraction of R&D personnel and are, on average, larger. More specifically, exporters are almost twice as productive as non-exporters and have software expenditures per employee that are almost three times as high as those reported by non-exporters, employ ten times as many R&D employees and are four times larger than non-exporters.



Export shares by exporter groups, Austrian manufacturing: 2002, 2004 and 2006



Exporter groups are defined as follows: 'marginal exporters' with export shares of equal to or less than 5 percent, 'low to medium exporters' with export shares of more than 5 but less or equal to 30 percent, 'high exporters' with export shares between 30 and 50 percent, and finally, 'very high exporters' with export shares exceeding 50 percent.

Source: own calculations

Table 3

Firm-level indicators, weighted means 2002-2006, total manufacturing (NACE 15-37)

Variable	Non-ex- porters	Exporters	Marginal exporters	Low to medium exporters	High ex- porters	Very high exporters
Wage/employee1	22.2	30.1	27.1	28.6	29.5	32.9
Labor productivity ¹	102.0	195.1	142.6	148.8	181.2	262.0
Software expenditures/employee1	0.085	0.224	0.209	0.174	0.211	0.276
R&D personnel/total employment	0.002	0.020	0.004	0.009	0.018	0.038
Size (total employment/firm)	32	132	62	76	119	215

Exporter groups are defined as follows: 'marginal exporters' with export shares of equal to or less than 5 percent, 'low to medium exporters' with export shares of more than 5 but less or equal to 30 percent, 'high exporters' with export shares between 30 and 50 percent, and finally, 'very high exporters' with export shares exceeding 50 percent.

¹ expressed in thousand EUR.

Source: own calculations

Furthermore, a comparison across exporter groups reveals that, in general, the performance improves from lower to higher exporter groups. The only exception is software expenditures per employee which turns out to be lowest for the second exporter group of 'low to medium exporters'. In general, the group of 'very high exporters' outperforms all other groups. The most dramatic differences are found for firm size and R&D efforts in that 'very high exporters' are almost four times larger and employ almost ten times as many R&D employees as 'marginal exporters'. This finding is consistent with predictions of the workhorse model in international trade (Melitz, 2003) which emphasizes that more exportoriented firms are more productive and profitable.⁵

7. Results

The following two sub-sections present empirical results on the prevalence and significance of rent-sharing in Austrian manufacturing and the role a firm's export status plays in its willingness to share part of its rents with its employees. More specifically, whether, exporters are willing to share a higher fraction of their rents with their workers is subject of subsection 7.1 while sub-section 7.2 looks at differences in the extent of rent-sharing across exporter groups with varying degrees of export-orientation, or equivalently, internationalization.

7.1 Exporters versus non-exporters

In Table 4 a random effects model with a one-year lag is estimated to account for unobserved firm heterogeneity. Column [1] in Table 4 shows that the wage bargaining process in Austrian manufacturing is indeed characterized by rent-sharing between employers and employees, even after controlling for observed and unobserved firm characteristics. Hence, in line with related studies, evidence is provided that employees are able to claim part of the rents or profits made by their employers.

Moreover, in line with recent empirical results, exporters are found to pay significantly higher wages on average, possibly in order to compensate for potentially riskier or more demanding activities carried out in exporting firms.

Furthermore, similar to evidence provided by Arai (2003) for Sweden, Fakhfakh and FitzRoy (2004) for France or Piekkola and Kauhanen (2003) for Finland, the analysis corroborates the existence of a significant size-premium in that larger firms indeed pay extra premia to their employees. Interestingly, the effect of firm-size only pushes through with a one-year lag.

⁵ More detailed descriptive statistics and econometric estimations of export premia are found in Stöllinger et al. (2010).

Table 4

Random effects model with 1-year lag exporters versus non-exporters

Variable	[1]	[2]
Constant	1.707***	1.503***
	(53.139)	(30.637)
Labor productivity t	0.244***	0.290***
	(34.301)	(26.413)
Labor productivity $t-1$	-0.001	-0.002
	(-0.208)	(-0.227)
Firm size t	0.002	0.003
	(0.164)	(0.271)
Firm size $_{t-1}$	0.058***	0.057***
	(6.136)	(6.038)
Software expenditures per employee t	0.006***	0.006***
	(4.335)	(4.334)
Software expenditures per employee $_{t-1}$	0.003*	0.002*
	(1.948)	(1.859)
R&D personnel/total employment t	0.093**	0.099**
	(2.186)	(2.325)
R&D personnel/total employment $_{t-1}$	0.174***	0.177***
	(3.894)	(3.963)
Export status (exporter=1)	0.022***	0.304***
	(2.864)	(5.845)
Export status*labor productivity		-0.060***
		(-5.479)
chi2	5967.243	6050.732
R-squared	0.599	0.604
No. of observations	6,236	6,236

The dependent variable is the log of nominal gross wage costs per employee. With the exception of R&D personnel/total employment, all variables are in logs. All estimations include 2-digit industry dummies and year dummies.

The estimations are based on a random-effects procedure with a one-year lag.

Absolute value of t-statistics in parentheses

* significant at 10%

** significant at 5%

*** significant at 1%

Extra premia are also paid by more skill or technology-intensive firms operating in Austrian manufacturing. Specifically, as highlighted by Korösi (2010) for a set of Hungarian firms or Piekkola and Kauhanen (2003) for Finnish firms, innovative Austrian firms which devote non-negligible resources to the development of product and/or process innovations also pay extra innovation premia to their employees. Hence, even though typically only a relatively small fraction of a firm's employees are involved in R&D activities, the entire workforce benefits from its employers' research efforts in terms of significantly higher wages. Interestingly, both skill-intensity and R&D efforts exert a rather persistent effect on workers' wages.

Column [2] in Table 4 furthermore highlights that, in contrast to non-exporters, exporters share a significantly smaller part of the rents they are able to realize with their workers. This is consistent with the idea that, in contrast to non-exporters, exporters can successfully exploit their threat-points of sourcing or producing abroad which weakens workers' bargaining position and reduces the share of the rents workers are able to cream off.

Table 5		
GMM estimat	ion with 1-year lag	
exporters ver	sus non-exporters	
Variable	[4]	[2]
Valiable	[']	[2]
Constant	0.020***	0.024***
Waraa	(10.230)	(6.096)
wages $t-1$	0.287***	0.285***
l shan analisiti itu	(7.666)	(7.665)
Labor productivity t	0.209***	0.225***
l shan analisiti itu	(17.549)	(11.868)
Labor productivity $t-1$	-0.085***	-0.084***
	(-5.855)	(-5.811)
Firm size _t	-0.025	-0.024
	(-1.564)	(-1.516)
Firm size $t-1$	0.035**	0.034**
	(2.511)	(2.481)
Software expenditures/employee t	0.005***	0.005***
	(3.132)	(3.130)
Software expenditures/employee $_{t-1}$	0.002	0.002
	(1.132)	(1.104)
R&D personnel/total employment t	0.045	0.045
	(0.744)	(0.753)
R&D personnel/total employment $_{t-1}$	0.063	0.063
	(0.939)	(0.933)
Export status (exporter=1)	0.035**	0.128
	(2.444)	(1.407)
Export status*labor productivity		-0.020
		(-1.043)
chi2	388.239	389.043
No. of observations	3,485	3,485
The dependent variable is the log of nominal gross wage of	osts per employee. With the exception of	R&D personnel/tota

The dependent variable is the log of nominal gross wage costs per employee. With the exception of R&D personnel/total employment, all variables are in logs. All estimations include 2-digit industry dummies and year dummies.

The estimations are based on the first-difference GMM estimation procedure as suggested by Arellano and Bond with a oneyear lag.

* significant at 10%

** significant at 5%

*** significant at 1%

Moreover, the generalized methods of moments estimation (GMM) as suggested by Arellano Bond (1991) is applied which generally gives more robust and efficient estimators. The procedure purges prevailing fixed effects by first-differencing and instruments the endogenous variable with its lagged values in levels.

Consistent with results presented in Table 4 above, column [1] in Table 5 again demonstrates that rent-sharing is an integral element of the wage-determination process in Austrian manufacturing. More specifically, the major effect on workers' wages actually stems from rent-sharing.

That exporters pay higher wages is again confirmed by the results. Hence, strong evidence of a positive export-premium is present.

Consistently, employees profit from working in larger firms by receiving significantly higher wages and salaries. A significant one-year lagged size-effect is again apparent. Moreover, more skill or technology-intensive Austrian firms pay extra premia to their employees, irrespective of skill-level. Hence, both low as well as high skilled employees profit from working in high-skilled, technology-intensive firms. In contrast to above results, however, effects are rather short-lived as only the current skill or technology-level matters for workers' wages. The overall effect on workers' wages is again rather small.

Finally, in stark contrast to the above findings of a positive innovation-premium, more innovative firms which seek to develop new products and/or processes are not found to pay significantly higher wages to their employees.

Column [2] in Table 5 also accounts for the potential role the threat-point of sourcing or producing abroad has for wage negotiations, and consequently, for rent-sharing between exporting firms and their employees. However, no evidence is found that exporters share a higher part of their rents with their workers which indicates that exporters are in general not able to exploit their threat points to induce their workers to concede to lower wages.

7.2 Across exporters with different exporter groups

Similar and comparable analyses are also carried out for exporters with different export shares as proxies for their degree of internationalization in order to establish whether above results also hold across exporter groups or whether systematic differences, particularly in terms of observed rent-sharing, can be observed across groups.

Table 6 presents results from random effects models with a one-year lag to account for unobserved firm heterogeneity. Column [1] again points at the central and pivotal role rent-sharing plays for the level of wages paid to employees in the Austrian manufacturing sector. Moreover, consistent with above results reported in Table 4, larger, more innovative and more skill or technology-intensive firms pay extra premia to their employees.

With respect to potential export premia, interesting differences emerge across exporter groups. Specifically, only the group of 'low to medium exporters' with export shares ranging between 5 to 30 percent pays significantly higher wages to their employees.

Table 6							
Random effects m	odel with 1-year lag						
Across exp	Across exporter groups						
Var	[1]	[2]					
Constant	1.701***	1.479***					
	(51.264)	(34.401)					
Labor productivity t	0.245***	0.308***					
	(34.239)	(33.523)					
Labor productivity $_{t-1}$	-0.001	0.001					
	(-0.129)	(0.099)					
Firm size t	0.002	0.006					
	(0.167)	(0.602)					
Firm size $t-1$	0.059***	0.054***					
	(6.252)	(5.790)					
Software expenditure/employee t	0.006***	0.005***					
	(4.332)	(4.212)					
Software expenditure/employee $_{t-1}$	0.003**	0.002*					
	(1.994)	(1.800)					
R&D personnel/total employment t	0.094**	0.105**					
	(2.209)	(2.477)					
R&D personnel/total employment $_{t-1}$	0.177***	0.180***					
	(3.965)	(4.071)					
Low to medium exporter (>5-30 %)	0.020***	0.306***					
	(2.695)	(5.424)					
High exporter (>30-50 %)	0.014	0.398***					
	(1.527)	(5.972)					
Very high exporter (>50 %)	0.009	0.668^^^					
	(0.908)	(11.002)					
Low to medium exporter*labor productivity		-0.060***					
Lich overstarticher productivity		(-5.261)					
High exponer labor productivity		-0.060					
Very high exporter*labor productivity		-0.133***					
		(-11.808)					
chi2	5958 561	6244 378					
R-squared	0.599	0.609					
N	6,236	6,236					

The dependent variable is the log of nominal gross wage costs per employee. With the exception of R&D personnel/total employment, all variables are in logs. All estimations include 2-digit industry dummies and year dummies.

The estimations are based on a random-effects procedure with a one-year lag.

Exporter groups are defined as follows: 'marginal exporters' with export shares of equal to or less than 5 percent, 'low to medium exporters' with export shares of more than 5 but less or equal to 30 percent, 'high exporters' with export shares between 30 and 50 percent, and finally, 'very high exporters' with export shares exceeding 50 percent.

* significant at 10%

** significant at 5%

*** significant at 1%

Column [2] in Table 6 also accounts for the potential role the threat-point of sourcing or producing abroad plays for the outcome of wage negotiations. Consistent evidence is found that Austrian manufacturing firms can effectively exploit their threat-points of sourcing or producing abroad so that, compared to their purely domestically oriented counterparts, they only share a significantly smaller part of their rents with their employees. But firms' ability to make use of their threat-points clearly differs and strongly hinges on the degree of their exposure to international markets. Specifically, more export-oriented firms can most credibly threaten to offshore or outsource activities which obviously puts them in a better bargaining position and therefore enables them to further cut down on the share of the rents their employees can claim.

Column [1] in Table 7 again applies a generalized methods of moment estimation (GMM) procedure and confirms that Austrian firms share part of the rents they realize with their employees. All in all, this rent-sharing effect appears to be dominant in the overall wage-determination process.

But interestingly, exporters pay significantly higher wages to their employees. Hence, employees profit from being employed by firms which actively and successfully supply foreign markets by receiving higher wages and salaries. However, the size of the export-premium employers are willing to pay varies with the degree of internationalization. Specifically, more export-oriented and international firms pay relatively lower export premia compared to firms whose returns from exporting only accounts for a relatively small fraction of their overall returns. Hence, employees benefit more from working in less international and export-oriented firms.

Whether the extent of rent-sharing systematically differs across exporter groups is subject of column [2] in Table 7. In contrast to above findings (see Table 6), the results demonstrate that *only* the group of 'very high exporters' with export shares exceeding 50 percent is able to take advantage of the threat-point effect of internationalization thereby inducing workers to accept lower wages.

Table 7

GMM estimation with 1-year lag across exporter groups

Var	[1]	[2]
Constant	0.020***	0.024***
	(10.300)	(6.274)
Wages _{t-1}	0.282***	0.275***
	(7.554)	(7.437)
Labor productivity t	0.211***	0.247***
	(17.683)	(14.096)
Labor productivity $_{t-1}$	-0.084***	-0.077***
	(-5.776)	(-5.347)
Firm size t	-0.025	-0.020
	(-1.581)	(-1.250)
Firm size $_{t-1}$	0.036***	0.036***
	(2.588)	(2.595)
Software expenditure/employee t	0.005***	0.005***
	(3.102)	(3.071)
Software expenditure/employee $t-1$	0.002	0.002
	(1.169)	(1.090)
R&D personnel/total employment t	0.046	0.048
	(0.774)	(0.801)
R&D personnel/total employment $_{t-1}$	0.067	0.069
	(1.005)	(1.038)
Low to medium exporter (>5-30 %)	0.035***	0.111
	(2.788)	(1.203)
High exporter (>30-50 %)	0.033**	0.153
	(1.984)	(1.368)
Very high exporter (>50 %)	0.022	0.375***
	(1.128)	(3.486)
Low to medium exporter*labor productivity		-0.016
		(-0.835)
High exporter*labor productivity		-0.024
		(-1.099)
very high exporter abor productivity		-0.069***
chi2	202 002	407 29F
	3 485	407.300
IN	3,400	3,403

The dependent variable is the log of nominal gross wage costs per employee. With the exception of R&D personnel/total employment, all variables are in logs. All estimations include 2-digit industry dummies and year dummies.

The estimations are based on the first-difference GMM estimation procedure as suggested by Arellano and Bond with a oneyear lag.

Exporter groups are defined as follows: 'marginal exporters' with export shares of equal to or less than 5 percent, 'low to medium exporters' with export shares of more than 5 but less or equal to 30 percent, 'high exporters' with export shares between 30 and 50 percent, and finally, 'very high exporters' with export shares exceeding 50 percent.

* significant at 10%

** significant at 5%

*** significant at 1%

8. Summary and conclusion

Whether globalization - an umbrella-term that captures the growing global integration of economies through trade, FDI, migration or the diffusion of technology – is a crucial determining factor during wage negotiations has aroused growing interest. Theoretically, it may lead to a redistribution of bargaining power between employers and employees, affect workers' and firms' threat-points or mold the size of profits firms are able to realize, therefore co-determining the wage employees are able to bargain for. And while interest for the subject is growing, evidence is still scarce and scattered though.

This paper contributes to the ongoing discussion and seeks to establish whether firms which operate internationally, more specifically, whether exporters share a higher part of their rents with their employees. The analysis focuses on Austria which is particularly suitable since Austria is a very open and outward oriented economy with a high share of firms exporting and importing tradable goods and services.

The analysis uses a panel of Austrian manufacturing firms for the period 2002 to 2006 and demonstrates that, in general, exporters do not share a higher fraction of their rents with their employees.

However, exporters are a very heterogeneous group, comprising firms with different degrees of export-orientation, different degrees to which they engage in international activities and different degrees to which they earn rents from their export activities. Hence, a more refined classification scheme was applied to account for within-group heterogeneity. The results highlight that some exporting firms are actually able to reduce the part of the rents they have to share with their workers. Specifically, the most international and exportoriented group of exporters with export shares above 50 percent shares a *smaller* fraction of its rents with its employees. This is consistent with the idea that more international firms are able to use their threat-points to their advantage as suggested by Kramarz (2003). Specifically, this particular group of exporters which is generally better informed about their outside options can more credibly threaten to outsource or offshore part of their production. This in turn induces employees to concede to more moderate or low wage rises and reduces the share of rents they are able to appropriate so as to avert the potential loss of employment.

Moreover, in line with other empirical results, evidence is found that other firmcharacteristics also matter for employees' pay. In that respect, larger firms consistently pay higher wages to their employees. Additionally, employees benefit from working in more skill or technology-intensive Austrian firms which also pay extra premia, irrespective of an employee's skill level. In contrast, however, more R&D intensive firms do not appear to pay significantly higher wages to their employees.

9. References

Abowd J.A. and T. Lemieux, 1993, The Effects of Product Market Competition on Collective Bargaining Agreements: The Case of Foreign Competition in Canada, The Quarterly Journal of Economics, 108(4), 983-1014.

Arai, M., 2003, Wages, Profits, and Capital Intensity: Evidence from Matched Worker-Firm Data, Journal of Labor Economics, 21(3), 593-618.

Arellano, M. and S. Bond, 1991, Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment equations, Review of Economic Studies, 58, 277-97.

Benito, A., 2000, Inter-Industry Wage Differentials in Great Britain, Oxford Bulletin of Economics and Statistics, 62(5), 727-46.

Bivens, J., 2006, Wages, Profits, and Rent-Sharing in an Open Economy, International Review of Applied Economics, 20(2), 69-83.

Bernard, A.B., J.B. Jensen and P.K. Schott, 2005, Importers, Exporters and Multinationals: A Portrait of Firms in the U.S. that Trade Goods, Yale School of Management Working Paper Series, Working Paper Series ES, Economics, Working Paper No. 39

Blanchflower, D.G., A.J. Oswald and P. Sanfey, 1996, Wages, Profits, and Rent Sharing, Quarterly Journal of Economics, 111(1), 227-51.

Brock, E. and S. Dobbelaere, 2006, Has International Trade Affected Workers' Bargaining Power, Review of World Economics, 142(2), 233-66.

Bronars, S. and M. Famulari, 2001, Shareholder Wealth and Wages: Evidence for White-Collar Workers, Journal of Political Economy, 109 (2), 328-54.

Choi, M., 2001, Threat Effect of Foreign Direct Investment on Labor Union Wage Premium, Political Economy Research Institute, Working Paper Series No. 27.

Christofides, L.N. and A.J. Oswald, 1992, Real Wage Determination and Rent-Sharing in Collective Bargaining Agreements, The Quarterly Journal of Economics, 107(3), 985-1002.

Dickens, W. and L. Katz, 1986, Inter-industry wage differences and industry characteristics, in: K. Lang and J. Leonard, (eds.), Unemployment and the structure of labor markets, London: Basil Blackwell.

Du Caju, P., E. Gautier, D. Momferatou and M. Ward-Warmeding, 2008, Institutional Features of Wage Bargaining in 23 European Countries, the US and Japan, IZA Discussion Paper No. 3867.

Fakhfakh, F. and F. FitzRoy, 2004, Basic Wages and Firm Characteristics: Rent Sharing in French Manufacturing, Labour, 18 (4), 615-31.

Goos, M. and J. Konings, 2000, Does Rent-Sharing Exist in Belgium? An Empirical Analysis Using Firm Level Data, Center for Economic Studies Discussion Paper Series 00.19.

Hildreth, A.J.K, 1998, Rent-Sharing and Wages: Product Demand or Technology Driven Premia?, Economics of Innovation and New Technology, 5, 199-226.

Hildreth. A. and A. Oswald, 1997, Rent-Sharing and Wages: Evidence from Company and Establishment Panels, Journal of Labor Economics, 15(2), 318-37.

International Monetary Fund, 2010, World Economic Outlook Database, April 2010.

Katz, L.F. and L.H. Summers, 1989, Industry Rents: Evidence and Implications, Brookings Papers on Economic Activity (Microeconomics), 209-90.

Knell, M. and A. Stiglbauer, 2009, Lohnstaffelung und Lohnführerschaft in Österreich – Bestandsaufnahme und Implikationen, Geldpolitik & Wirtschaft, 04/09.

Korösi, G., 2010, Innovation and Rent Sharing in Corporate Wage Setting in Hungary, Micro-Dyn Working Paper 09/10.

Kramarz, F., 2003, Wages and International Trade, CEPR Discussion Paper No. 3936.

Kouwenberg, J. and van Opstal, R., 1999, Inter-industry Wage Differentials: Evidence from Micro Data, CPB Report 1999/3, Quarterly Review of CPB Netherlands Bureau for Economic Policy Analysis.

Lester, R., 1952, A Range Theory of Wage Differentials, Industrial and Labour Relations Review, 5(4), 483-500.

Margolis, D. and K. Salvanes, 2001, Do Firms Really Share Rents with Their Workers?, IZA DP No. 330.

Martin, P.S., 2003, Rent Sharing Before and After the Wage Bill, Mimeo, Department of Economics, University of Warwick.

Martins, P.S. and L.D. Opromolla, 2010, Exports, Imports and Wages: Evidence from Matched Firm-Worker-Product Panels, Estudos e Documentos de Trabalho, Working Papers 2, 2010.

Melitz, M.J., 2003, The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity, Econometrica, 71(6), 1695-1725.

Munch, J.R. and J.R. Skaksen, 2008, Human Capital and Wages in Exporting Firms, Journal of International Economics, 75 (2), 363–72.

OECD, 2004, Wage-Setting Institutions and Outcomes. In: OECD Employment Outlook 2004, Chapter 3, 127–181.

Piekkola, H. and A. Kauhanen, 2003, Rent-sharing as Firm-Level Pay, International Journal of Manpower, 24 (4), 426-51.

Rodrik, D., 1999, Globalisation and Labour, or: if globalisation is a bowl of cherries, why are there so many glum faces around the table? in R. Baldwin et al. (eds.), Market Integration, Regionalism and the Global Economy, Cambridge: Cambridge University Press.

Schank, T., Schnabel, C. and J. Wagner, 2007, Do exporters really pay higher wages? First evidence from German linked employer-employee data, Journal of International Economics, 72 (1), 52–72.

Slichter, S., 1950, Notes on the Structure of Wages, Review of Economics and Statistics, 32, 80-91.

Stöllinger R., R. Stehrer and J. Pöschl, 2010, Austrian exporters: A firm-level analysis, Micro-Dyn Working Paper No. 18/10.

Tsou, M-W., J-T Liu and C.J. Huang, 2006, Export Activity, Firm Size and Wage Structure: Evidence from Taiwanese Manufacturing Firms, Asian Economic Journal, 20(4), 333-54.

Wirtschaftskammer Österreich (WKÖ), 2009, Streiks in Österreich

10. Appendix

Table 8									
	Maj	or trading p	artners	for Austria	n imports	s: import s	hares, 19	80-2008	
Year	EU	Germany	Italy	Switzer- land	Nether- lands	Czech Republic	France	China	USA
1980	68.48	40.85	9.10	5.00	2.70		3.95	0.67	3.38
1985	68.27	41.00	8.24	4.54	2.65		3.63	0.71	3.68
1990	73.56	43.89	9.01	4.51	2.82		4.19	1.27	3.64
1995	78.89	43.59	8.76	3.89	3.43	1.89	4.91	1.64	4.24
2000	80.19	43.78	6.84	4.74	4.30	2.47	3.76	1.21	4.08
2001	80.52	43.43	6.71	4.70	4.56	2.60	3.81	1.66	3.84
2002	80.80	42.58	6.62	4.79	4.44	2.79	3.68	1.52	3.48
2003	81.85	43.36	6.72	5.04	4.24	3.08	3.78	1.87	2.23
2004	82.74	46.33	6.82	4.29	3.88	3.05	3.85	1.66	1.93
2005	80.50	45.84	6.64	4.52	3.75	3.08	3.76	2.15	2.23
2006	79.65	45.44	6.98	4.48	4.09	3.05	3.16	2.37	2.24
2007	79.31	45.51	7.09	4.99	4.32	3.07	3.12	2.54	2.22
2008	78.10	44.47	7.07	5.17	4.13	3.42	3.07	2.61	1.79

Note: The countries are ranked according to prevailing import shares in 2008.

Source: Direction of Trade Statistics, IMF

Table 9

Major trading partners for Austrian exports: export shares, 1980-2008

Year	FU	Germany	Italy	USA	Switzer-	Czech Republic	France	Hungary	uк
. oui		Connuny	itary	00/1	land	Republic	i i unico	nungury	•
1980	66.50	30.90	10.98	2.18	7.54		3.47	2.18	3.67
1985	63.80	30.13	8.95	4.63	6.76		3.96	2.58	4.60
1990	71.50	37.53	9.73	3.19	7.18		4.71	2.25	3.84
1995	77.18	38.35	8.87	2.97	5.85	2.74	4.44	3.64	3.30
2000	74.78	33.33	8.83	4.98	6.72	2.76	4.47	4.99	4.30
2001	74.88	32.33	8.69	5.14	5.42	2.77	4.94	4.55	4.79
2002	74.95	31.47	9.31	4.95	5.39	2.73	4.65	4.32	4.94
2003	75.27	31.51	9.70	4.89	5.18	2.87	4.72	3.98	4.74
2004	73.55	31.96	8.87	5.97	4.79	2.94	4.24	3.65	4.24
2005	71.83	31.06	8.65	5.82	5.14	2.94	4.14	3.33	3.96
2006	71.99	30.08	9.00	5.79	4.69	3.17	3.76	3.41	3.80
2007	72.55	29.80	8.84	4.90	4.33	3.57	3.59	3.65	3.57
2008	72.24	29.51	8.57	4.29	4.17	3.95	3.71	3.68	3.16

Note: The countries are ranked according to prevailing export shares in 2008.

Source: Direction of Trade Statistics, IMF

Short list of the most recent wiiw publications (as of February 2011

For current updates and summaries see also wiiw's website at <u>www.wiiw.ac.at</u>

Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms

by Sandra M. Leitner and Robert Stehrer

wiiw Working Papers, No. 73, February 2011 28 pages including 9 Tables and 2 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Change Begets Change: Employment Effects of Technological and Non Technological Innovations – a Comparison across Countries

by Sandra M. Leitner, Johannes Pöschl and Robert Stehrer

wiiw Working Papers, No. 72, January 2011 28 pages including 3 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 12/10

edited by Leon Podkaminer

- A structural decomposition of international trade
- Migration, skills and productivity
- · Structural root causes of the debt crises/heterogeneity
- Statistical Annex: Selected monthly data on the economic situation in Central and Eastern Europe

wiiw, December 2010

22 pages including 10 Tables and 3 Figures

(exclusively for subscribers to the wiiw Service Package)

wiiw Monthly Report 11/10

edited by Leon Podkaminer

- Albania: agnostic growth forecast
- · Bosnia and Herzegovina: economic recovery and new political constellation
- · Croatia: recovery not yet in sight
- · Kazakhstan: rebound is faster
- · Macedonia: slow and stable
- · Montenegro: recovery delayed
- · Russian Federation: oil-fuelled recovery loses steam
- · Serbia: inflationary pressures
- Turkey: swift return to pre-crisis levels
- · Ukraine: the return of high inflation
- Statistical Annex: Selected monthly data on the economic situation in Southeast Europe, Russia and Ukraine

wiiw, November 2010

38 pages including 20 Tables

(exclusively for subscribers to the wiiw Service Package)

The role of services in the New Member States: A comparative analysis based on inputoutput tables

by Doris Hanzl-Weiss and Robert Stehrer

wiiw Research Reports, No. 366,November 2010 91 pages including 26 Tables and 37 Figures hardcopy: EUR 24.00 (PDF: EUR 15.00)

Migration, Skills and Productivity

by Peter Huber, Michael Landesmann, Catherine Robinson and Robert Stehrer with Robert Hierländer, Anna Iara, Mary O'Mahony, Klaus Nowotny and Fei Peng

> wiiw Research Reports, No. 365,November 2010 172 pages including 81 Tables and 20 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Inequality, Growth and Public Spending in Central, East and Southeast Europe

by Mario Holzner

wiiw Working Papers, No. 71, October 2010 15 pages including 3 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 10/10

edited by Leon Podkaminer

- · Bulgaria: muddling through
- Czech Republic: changing structure of growth
- Hungary: slow recovery based on external demand
- · Poland: waiting for the investment take-off
- Romania: recovery postponed
- · Slovakia: robust exports prevailing over weak domestic demand
- · Slovenia: catching up slowly
- · Baltics: : exports and restocking drive growth revival
- Statistical Annex: Selected monthly data on the economic situation in Central and Eastern Europe

wiiw, Octoberr 2010 42 pages including 17 Tables

(exclusively for subscribers to the wiiw Service Package)

The Impact of Preferential Trade Agreements on the Margins of International Trade

by Neil Foster, Johannes Pöschl and Robert Stehrer

wiiw Working Papers, No. 70, September 2010 25 pages including 9 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Discrepancies between Purchasing Power Parities and Exchange Rates under the Law of One Price: A Puzzle (partly) Explained?

by Leon Podkaminer

wiiw Working Papers, No. 69, September 2010 18 pages including 5 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 8-9/10

edited by Leon Podkaminer

- The intensity of competition in Central, Eastern and Southeast Europe: stylized facts and repercussions for sectoral price developments
- Exchange market pressure contagion in CESEE
- Scenarios for Ukraine's medium- and long-term development
- Markets and morals
- Statistical Annex: Selected monthly data on the economic situation in Central and Eastern Europe

wiiw, August-September 2010

36 pages including 10 Tables and 7 Figures

(exclusively for subscribers to the wiiw Service Package)

An Empirical Characterization of Redistribution Shocks and Output Dynamics

by Klemens Hauzenberger and Robert Stehrer

wiiw Working Papers, No. 68, August 2010 47 pages including 6 Tables and 4 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Ukraine, the European Union and the International Community: Current Challenges and the Agenda for Overcoming the Stalemate

by Vasily Astrov, Igor Burakovsky, Grzegorz Gromadzki, Peter Havlik and Vasyl Yurchyshyn

wiiw Research Reports, No. 364, July 2010 93 pages including 27 Tables and 15 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Austrian Exporters: A Firm-Level Analysis

by Roman Stöllinger, Robert Stehrer and Johannes Pöschl

wiiw Working Papers, No. 67, July 2010 32 pages including 14 Tables and 4 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Will Exports Prevail over Austerity?

by Vasily Astrov, Mario Holzner, Kazimierz Laski, Leon Podkaminer et al.

wiiw Current Analyses and Forecasts. Economic Prospects for Central, East and Southeast Europe, No. 6, July 2010
164 pages including 30 Tables and 33 Figures
hardcopy: EUR 80.00 (PDF: EUR 65.00)

wiiw Monthly Report 7/10

edited by Leon Podkaminer

- A few reflections on fiscal and monetary policy in the euro area
- Unit labour costs, exchange rates and responses to the crisis
- Poland's presidential election: and the winner is ...
- Statistical Annex: Selected monthly data on the economic situation in Southeast Europe, Russia and Ukraine

wiiw, July 2010

28 pages including 9 Tables and 6 Figures

(exclusively for subscribers to the wiiw Service Package)

wiiw Service Package

The Vienna Institute offers to firms and institutions interested in unbiased and up-to-date information on Central, East and Southeast European markets a package of exclusive services and preferential access to its publications and research findings, on the basis of a subscription at an annual fee of EUR 2,000.

This subscription fee entitles to the following package of **Special Services**:

- A free invitation to the Vienna Institute's Spring Seminar, a whole-day event at the end of March, devoted to compelling topics in the economic transformation of the Central and East European region (for subscribers to the wiiw Service Package only).
- Copies of, or online access to, *The Vienna Institute Monthly Report*, a periodical consisting of timely articles summarizing and interpreting the latest economic developments in Central and Eastern Europe and the former Soviet Union. The statistical annex to each *Monthly Report* contains, alternately, country-specific tables or graphs with monthly key economic indicators, economic forecasts, the latest data from the wiiw Industrial Database and excerpts from the wiiw FDI Database. This periodical is not for sale, it can only be obtained in the framework of the wiiw Service Package.
- Free copies of the Institute's **Research Reports** (including **Reprints**), **Current Analyses and Forecasts**, **Country Profiles** and **Statistical Reports**.
- A free copy of the *wiiw Handbook of Statistics* (published in October/November each year and containing more than 400 tables and graphs on the economies of Albania, Bosnia and Herzegovina, Bulgaria, Croatia, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Macedonia, Montenegro, Poland, Romania, Russia, Serbia, the Slovak Republic, Slovenia and Ukraine)
- Free online access to the wiiw Monthly Database, containing more than 1200 leading indicators monitoring the latest key economic developments in ten Central and East European countries.
- Consulting. The Vienna Institute is pleased to advise subscribers on questions concerning the East European economies or East-West economic relations if the required background research has already been undertaken by the Institute. We regret we have to charge extra for *ad hoc* research.
- Free access to the Institute's specialized economics library and documentation facilities.

Subscribers who wish to purchase wiiw data sets **on CD-ROM** or special publications not included in the wiiw Service Package are granted considerable **price reductions**.

For detailed information about the wiiw Service Package please visit wiiw's website at www.wiiw.ac.at

To The Vienna Institute for International Economic Studies Rahlgasse 3 A-1060 Vienna

- O Please forward more detailed information about the Vienna Institute's Service Package
- O Please forward a complete list of the Vienna Institute's publications to the following address

Please enter me for

- 1 yearly subscription of *Research Reports* (including *Reprints*) at a price of EUR 120.00 (hardcopy, Austria),
 EUR 135.00 (hardcopy, Europe), EUR 155.00 (hardcopy, overseas) and EUR 48.00 (PDF download with password) respectively
- 1 yearly subscription of *Current Analyses and Forecasts* a price of EUR 150.00 (hardcopy, Austria),
 EUR 155.00 (hardcopy, Europe), EUR 170.00 (hardcopy, overseas) and EUR 120.00 (PDF download with password) respectively

Please forward

0	the following issue of Research Reports
0	the following issue of Current Analyses and Forecasts
0	the following issue of Working Papers
0	the following issue of Research Papers in German language
0	the following issue of wiiw Database on Foreign Direct Investment
0	the following issue of wiiw Handbook of Statistics
0	(other)

Name		
Address		
Telephone	Fax	E-mail
Date		Signature

Herausgeber, Verleger,	Eigentümer und Hersteller:
	Verein "Wiener Institut für Internationale Wirtschaftsvergleiche" (wiiw), Wien 6, Rahlgasse 3
	ZVR-Zahl: 329995655
Postanschrift:	A-1060 Wien, Rahlgasse 3, Tel: [+431] 533 66 10, Telefax: [+431] 533 66 10 50
Internet Homepage:	www.wiiw.ac.at
Nachdruck nur auszug	sweise und mit genauer Quellenangabe gestattet.
P.b.b. Verlagspostamt	1060 Wien